Author's response to reviews

Title: A double blind randomized placebo control crossover trial on the effect of dietary nitrate supplementation on exercise tolerance in stable moderate chronic obstructive pulmonary disease

Authors:

Paul Leong (paul.leong@monashhealth.org)
Jane E Basham (janebasham@hotmail.com)
Theresa Yong (theresa.yong@monashhealth.org)
Adrian Chazan (adrianchazan@gmail.com)
Paul Finlay (paul.finlay@monashhealth.org)
Sara Barnes (sara.barnes@monashhealth.org)
Philip G Bardin (philip.bardin@monash.edu)
Donald Campbell (donald.campbell@monash.edu)

Version: 5
Date: 20 April 2015

Author's response to reviews: see over
Editorial requests

1. Include date of trial registration.
   a. Added under Methods in main text, third line of that section

2. Abbreviation section: Please provide a list of abbreviations used in your manuscript, if any.
   a. Added: see after Conclusions section and preceding Competing Interests

3. Include the funding information in the Acknowledgement Section.
   a. Acknowledgement and Funding are no longer separate; funding subheading deleted and content incorporated under Acknowledgment.